Author's response to reviews

Title: 3066 consecutive Gamma Nails. 12 years experience at a single centre.

Authors:

Alicja J Bojan (alicja.bojan@vgregion.se)
Claudia Beimel (claudia.beimel@stryker.com)
Andreas Speitling (andreas.speitling@stryker.com)
Gilbert Taglang (gtaglang@wanadoo.fr)
Carl Ekholm (carl.ekholm@vgregion.se)
Anders Joensson (anders.joensson@stryker.com)

Version: 7 Date: 24 June 2010

Author's response to reviews: see over
RE: 3066 consecutive Gamma Nails. 12 years experience at a single centre

Alicja J. Bojan¹, Claudia Beimel ², Andreas Speitling ², Gilbert Taglang³, Carl Ekholm¹, Anders Jönsson¹,².

¹ Department of Orthopaedics, Institute of Clinical Sciences at Sahlgrenska Academy, Gothenburg University, Gothenburg, Sweden
² Stryker Osteosynthesis, Schönkirchen, Germany
³ Trauma Unit University Hospital of Strasbourg, France

Dear Editors,

We have corrected the editorial revisions as following:
  1. The numbering of tables has been corrected; Table 1 reappears also in the Methods part after Table 2 and 3.
  2. The statement “All authors read and approved the final manuscript.” has been included in the Authors’ contributions part.
  3. The Figures 1 and 2 have been cropped.

Yours sincerely,

Alicja Bojan, MD